Title: Indications for computed tomography (CT-) diagnostics in proximal humeral fractures: a comparative study of plain radiography and computed tomography

Authors:

Christian Bahrs (C.Bahrs@gmx.de)  
Bernd Rolauffs (berndrolauffs@yahoo.de)  
Norbert P. Südkamp (norbert.suedkamp@uniklinik-freiburg.de)  
Hagen Schmal (hagen.schmal@uniklinik-freiburg.de)  
Christoph Eingartner (Christoph.Eingartner@ckbm.de)  
Klaus Dietz (klaus.dietz@uni-tuebingen.de)  
Philippe L. Pereira (philippe.pereira@med.uni-tuebingen.de)  
Kuno Weise (kweise@bgu-tuebingen.de)  
Erich Lingenfelter (erichlingenfelter@yahoo.com)  
Peter Helwig (peter.helwig@uniklinik-freiburg.de)

Version: 6 Date: 7 March 2009

Author's response to reviews:

Dear ladies and gentlemen,

According to the formatting changes that are requested, we changed the reference 17 from


to


We changed 2-D CT and 3-D CT to 2 D CT and 3 D CT in the text.

We did not changed the abstract since it is correct according to structure and count of words.

Sincerely

C.Bahrs